Author’s response to reviews

Title: Consensus on treatment for residents in long-term care facilities: perspectives from relatives and care staff in the PACE cross-sectional study in 6 European countries.

Authors:

Maud ten Koppel (m.tenkoppel@vumc.nl)
H. Roeline Pasman (hrw.pasman@vumc.nl)
Jenny van der Steen (J.vanderSteen@vumc.nl)
Hein van Hout (hpj.vanhout@vumc.nl)
Marika Kylänen (marika.kylanen@thl.fi)
Lieve Van den Block (Lieve.Van.den.Block@vub.be)
Tinne Smets (Tinne.Smets@vub.be)
Luc Deliens (Luc.Deliens@vub.be)
Giovanni Gambassi (Giovanni.Gambassi@unicatt.it)
Katherine Froggatt (k.froggatt@lancaster.ac.uk)
Katarzyna Szczerońska (katarzyna.szczerońska@uj.edu.pl)
Bregje Onwuteaka-Philipsen (b.philipsen@vumc.nl)

Version: 2 Date: 27 Feb 2019

Author’s response to reviews:

Response to the reviewers

Dear dr. Foley,

We have included a clean version of our manuscript, in which we have adapted the reference list in such way that it adheres to the BMC referencing style, including abbreviations to all referred journals.
Yours sincerely,

Maud ten Koppel, on behalf of, Roeline Pasman, Jenny van der Steen, Hein van Hout, Marika Kylänen, Lieve Van den Block, Tinne Smets, Luc Deliens, Giovanni Gambassi, Katherine Froggatt, Katarzyna Szczerbińska and Bregje Onwuteaka-Philipsen.